Author's response to reviews

Title: Brain imaging in Kufs disease type B. Case report

Authors:

- Roberto Di Fabio (rodifa@gmail.com)
- Claudio Colonnese (claudio.colonnese@uniroma1.it)
- Filippo M Santorelli (filippo3364@gmail.com)
- Liliana Pestillo (pestilloliliana@virgilio.it)
- Francesco Pierelli (francesco.pierelli@uniroma1.it)

Version: 3 Date: 14 May 2015

Author's response to reviews:

To Prof Timothy Shipley 14.5.2015
Editor-in-Chief of BMC Neurology

Dear Editor,

Please find enclosed the revised version of the manuscript entitled “Brain imaging in Kufs disease type B. Case report” that we submit for a possible publication in the BMC Neurology as case report.

Line numbering and page numbers were added as well as the e-mail addresses of all authors.

Yours sincerely,

Roberto Di Fabio, MD
Department of Medical-Surgical Sciences and Biotechnologies
Sapienza University of Rome
Corso della Repubblica, 79
04100 – Latina, Italy
Phone +39 0773 6513327
rodifa@gmail.com